Author's response to reviews

Title: Randomized double-blind placebo-controlled trial of sublingual immunotherapy in children with house dust mite allergy in primary care: study design and recruitment

Authors:

Cindy M.A. de Bot (c.debot@erasusmc.nl)
Heleen Moed (h.moed@erasusmc.nl)
Marjolein Y Berger (m.berger@erasusmc.nl)
Esther Röder (e.roder@erasusmc.nl)
Hans de Groot (grooth@rdgg.nl)
Johan C de Jongste (j.c.dejongste@erasusmc.nl)
Roy Gerth van Wijk (r.gerthvanwijk@erasusmc.nl)
Johannes C van der Wouden (j.vanderwouden@erasusmc.nl)

Version: 7 Date: 20 October 2008

Author's response to reviews:

Date: October 20th 2008
Subject: Resubmission manuscript MS: 2623587981961243
Randomized double-blind placebo-controlled trial of sublingual immunotherapy in children with house dust mite allergy in primary care: study design and recruitment

Dear Editor,

This is the revised final version of the manuscript.

On behalf of all authors,
Sincerely yours,

Cindy de Bot
Erasmus MC Rotterdam
The Netherlands